Author's response to reviews

Title: Under the banyan tree - exclusion and inclusion of people with mental disorders in rural North India

Authors:

Kaaren Mathias (kaarenmathias@gmail.com)
Michelle Kermode (mkermode@unimelb.edu.au)
Miguel San Sebastian (miguel.sansebastian@epiph.umu.se)
Mirja Koschorke (mirja.koschorke@kcl.ac.uk)
Isabel Goicolea (isabel.goicolea@epiph.umu.se)

Version: 2
Date: 20 January 2015

Author's response to reviews:

Dear Editor

I have made the following changes

1. Added author email addresses to title page
2. Structured the abstract into sections
3. Inserted page numbers and line numbers into the main manuscript

Yours sincerely

Kaaren Mathias